

Measuring patient activation: the utility of the Patient Activation Measure administered in an interview setting

 $\label{eq:magdalena} Magdalena \ Holter^1 \cdot Alexander \ Avian^{1} \textcircled{0} \cdot Martin \ Weger^2 \cdot Sanja \ Strini^2 \cdot Monja \ Michelitsch^2 \cdot Katja \ Brenk-Franz^3 \cdot Andreas \ Wedrich^2 \cdot Andrea \ Berghold^1$

Accepted: 23 January 2024 / Published online: 22 February 2024 © The Author(s) 2024

Abstract

Background Patient activation is an emerging field in healthcare research concerning knowledge, skills, and confidence of patients in managing their health. This is particularly important for patients with chronic diseases, who often require more complex care management and self-care skills. However, due to temporary or longer-lasting visual impairments, certain patient groups cannot answer a questionnaire independently. The main objective is to investigate the psychometric properties of the German Patient Activation Measure® (PAM) survey in an everyday clinical setting where it has to be read aloud. **Methods** Outpatients with macular edema participated in this questionnaire-based cross-sectional study. The study assessed patient activation by the PAM® survey, self-rated health, self-efficacy, quality of life, and general mood. Interviewers read questionnaires aloud to patients. Psychometric properties of the PAM® survey were investigated by item response theory (IRT), Cronbach's α and trait–trait correlations.

Results The analysis included N = 554 patients. Median age was 69 (IQR 62.0–76.0) years and mean overall activation score 74.1 (SD 13.7). All items showed ceiling effects. Empirical reliability from the IRT model and Cronbach's α were 0.75. The PAM® survey showed a Spearman correlation of 0.54 with self-efficacy, 0.51 with quality of life and 0.34 with general mood. **Conclusion** The read-aloud PAM® survey has been shown to provide to adequate measurement precision and convergent validity to be used as a screening tool in an everyday clinical setting. Objective assessment in an interview setting with the PAM® survey is possible. PAM® items are good in distinguishing lower to middle activated patients, but not patients with high activation. Further, issues with structural validity need more investigation.

Keywords Patient Activation Measure® · Psychometrics · Item response theory · Mode of questionnaire administration

Abbreviations		DIF	Differential item functioning	
AIC _c	Akaike information criterion corrected	GPCM	General partial credit model	
BMI	Body mass index	IRT	Item response theory	
CFI	Comparative fit index	NCDIF	Non-compensatory differential item	
CTT	Classical test theory		functioning	
Alexander Avian alexander.avian@medunigraz.at		Andreas Wedrich andreas.wedrich@medunigraz.at		
Magdalena Holter magdalena.holter@medunigraz.at		Andrea Berghold andrea.berghold@medunigraz.at		
Martin Weger martin.weger@medunigraz.at Sanja Strini		¹ Institute for Medical Informatics, Statistics and Documentation, Medical University of Graz, G		
		Aust	a	
sanja.st	rini@medunigraz.at	² Depa	tment of Ophthalmology, Medical University of Graz Austria	
Monja I monia r	Michelitsch nichelitsch@medunigraz.at	Graz		
Katja Brenk-Franz katja.brenk-franz@med.uni-jena.de		³ Instit and I Gern	ute of Psychosocial Medicine, Psychotherapy Psychooncology, Jena University Hospital, Jena, nany	

RMSEA	Root mean square error of approximation
SABIC	Sample adjusted Bayesian information criterion
SRMSR	Standardized root mean square residual
TLI	Tucker–Lewis index
PAM®	Patient activation measure

Plain English Summary

Patient activation, the patients' knowledge, skills, and confidence in managing their disease is crucial for better health outcomes. The Patient Activation Measure (PAM®) is used in different healthcare settings to gain insight about patient's activation level. However, it is unclear if the PAM® works as intended with patients suffering from a chronic disease affecting the eye, unable to read and fill the questionnaire by themselves. In everyday clinical settings, we used the PAM® survey with patients suffering from macular edema, ensuring accessibility by reading it aloud. This study indicates that objective assessment in an interview setting with the PAM survey is possible. The PAM® survey showed to sufficiently measure patient activation if read aloud. It can be made accessible to many patients as it can be used in interviews. The PAM® survey quickly identifies patients with low patient activation.

Background

The number of adult patients suffering from a chronic disease e.g. diabetes or hypertension is steadily rising worldwide [1]. Those patients have to deal with complex treatments, may have to change their lifestyle and should be aware when their health status may deteriorate. To provide a better health outcome, patients suffering from a chronic disease should engage in their health care [2]. Therefore, patient activation is essential [2]. It describes knowledge, skills, and confidence of patients in managing their own health. More activated adult patients use more self-management services, show better self-management behavior, and are more compliant in taking their medication [3]. In addition, they believe in being in control of their own health [4]. Highly activated patients show more self-efficacy [5-7], higher quality of life [3, 8, 9] and lower depression scores [10, 11]. Patient activation can be increased through interventions, such as patient education or self-management coaching [12].

Patient activation can be assessed by the patient activation measure (PAM® survey), a widely used self-report questionnaire [3, 4, 6–9, 13]. It consists of 13 statements regarding ones' health behavior and attitudes towards health management. By the PAM® survey, a patient can be categorized into one of four levels of patient activation. At level 1, patients are not engaged in managing their own health; at level 2, they lack confidence to take action to maintain and improve their health; at level 3, they take action by positive changes in their health behaviors; and at level 4, they maintain a healthy lifestyle even during stressful times. Reliability and validity of the questionnaire were endorsed by multiple studies e.g. [2, 13]. The PAM® survey was translated into several languages [14, 15] including German [5] and completed by various patient groups (e.g., mental problems, neurological diseases, multimorbid patients [13]). Patients with wet age-related macular degeneration showed high activation [16]. The activation scores of patients with vascular diseases were in a normal range [17]. For patients with diabetes slightly lower scores were found [18, 19]. Overall, the PAM® survey seems to be a reliable and valid measurement tool for assessing patient activation in multiple languages and different patient groups.

Nonetheless some patient groups are not able to complete patient-reported outcomes on their own, e.g., because of impaired vision. For those patient's questionnaire can be made accessible in an interview setting. However, since most questionnaire are developed for self-administration, completing them in an interview could potentially lead to different results [20]. The self-administered paper questionnaire is visual, while the interview setting is aural, which can evoke different cognitive processes and therefore different responses [20]. Moreover, the interaction between the interviewer and participant can vary and the issue of interviewer bias arises [21]. This describes the unconsciously reaction of an interviewer in a way that prompts the participant to provide a response that is biased toward the interviewer's reactions [21]. Additionally, during interviews participants provided more positive and socially acceptable answers than during self-administration [20].

The PAM® survey was self-administered and applied in an interview setting [22]. From a psychometric perspective it is of interest how the PAM® survey works in case it is read aloud, which is especially relevant for patients with a chronic disease affecting vision and not known yet. The main goal of this study was to gain insight into how the PAM® survey can be used in an everyday clinical setting where it needs to be read aloud for patients.

Methods

Study design

This is a questionnaire-based cross-sectional study. It consists of an ad hoc sample from the population of outpatients from the Department of Ophthalmology of the Medical University of Graz. Data collection was performed from March 2020 until the end of February 2022.

The ethical committee of the Medical University of Graz approved the study (32-101 ex 19/20).

Participants

A macular edema is an accumulation of fluid in the macula impairing vision and can lead to severe visual impairment. It is a heterogeneous disease with different possible etiologies. This study focuses on two different etiologies: diabetes and retinal vein occlusion, the most common retinal vascular diseases [23]. Diabetic macular edema represents the most prevalent subtype, affecting an estimated 5.5% of the global diabetic population [24]. While diabetic macular edema develops slowly and occurs mostly bilateral, the onset of retinal vein occlusion is sudden and most frequently only one eye is affected. Most patients with a macular edema are treated with intravitreal injections, which means monthly visits at the hospital. Macular edemas due to diabetes and retinal vein occlusion can also be treated by patients' health behavior. Both types of macular edema have similar risk factors and are due to vascular risk factors.

This study enrolled patients with center-involving macular edema resulting from diabetes or retinal vein occlusion. They had to be aged 18 or older, speak German well enough for questionnaire comprehension, and possess hearing abilities for verbal communication. Exclusion criteria involved cognitive impairment.

Data collection

After signing informed consent, each patient enrolled was asked to complete the questionnaires once. This took place in-between medical routine eye-examinations. Because of these examinations patients could not read questionnaires by themselves. Thus, questionnaires were read out loud by one of four trained interviewers (MH, VW, AK, JG) using a standard procedure to guarantee objectivity and comparability of each interview. The training included practicing the interview procedure and familiarizing with outcome measures. Possible barriers and difficulties encountered during interviews were discussed, and appropriate behavior rehearsed. For instance, guiding patients to respond using a scale rather than providing a narrative or story was practiced, without influencing their answers towards any particular category. Possible answers were printed out in big font and located in front of patients. Participants were pseudonymized with the web-based pseudonymization tool 'iPSN' [25]. LimeSurvey [26] was used to gather and store answers of participants with their pseudonym.

Outcome measures

The PAM® survey [13] assesses knowledge, skills, and confidence of patients in managing their own health. The items are rated on a Likert-type scale with five response categories from "Disagree strongly" to "Agree strongly" and "Not applicable". Answers to the 13 items are summed up and transformed to a scale between 0 and 100 [13]. The German version of the PAM® survey (PAM-13D) [5] has been a reliable and valid questionnaire, showing a Cronbach's α of 0.84, factorial structure and a trait–trait correlation of r=0.43 between the score of the PAM-13D and general selfefficacy [5].

Furthermore, the trait well-being inventory mood level scale [27] was used to measure general mood and general quality of life. Cronbach's α is 0.83 for the scale general mood and 0.87 for the scale general quality of life [27]. Quality of life was conceptualized as an overall measure of life satisfaction, focusing on the cognitive dimension of subjective well-being encompassing beliefs about the present, past, and future, rather than specific aspects of life satisfaction in distinct domains. Strong positive correlations with life-satisfaction indicated construct validity [27]. Moreover, we assessed subjective belief to successfully cope with new demanding situations by own strength by the general self-efficacy scale [28]. In several German samples, a Cronbach's α between 0.80 and 0.90 was found [28]. Validity is given by correlations of self-efficacy with various other constructs, such as negative correlations with depression, anxiety, and burnout. For these questionnaires used in the study, mean score over item answers were built. Additionally, self-perceived health status was rated in five categories ("Very bad", "Bad", "Moderate", "Good", "Very good") [29]. Moreover, demographic data like sex, age, and education were assessed. Net income was investigated in five categories, corresponding to the income quintiles for elderly in Austria (www.statistik.at).

Data analysis

To achieve the study objective of investigating the psychometric properties of the PAM, a minimum of 500 patients were required to obtain stable estimates [30], further information are included in the Supplementary Files.

Categorical data are presented as absolute and relative frequencies, continuous data as means and standard deviations or medians and interquartile ranges, as appropriate. To gain the PAM® score, the answer category "Not applicable" was transformed into missing values and the raw scores of the PAM-13D were summed up and transferred to a scale between 0 and 100, according to the algorithm by Insignia Health, the company licensing the questionnaire (https:// www.insigniahealth.com/products/pam). To analyze the PAM-13D, we used item response theory (IRT), a set of statistical models describing the relationship between questionnaire items and person ability. Ability is observed through the answers given to questionnaire items. With higher person ability, higher categories are chosen (e.g., "Agree strongly"). Firstly, assumptions of IRT analysis were examined. The generalized partial credit model (GPCM) was chosen out of different IRT models based on fit-indices, LR-tests and Vuong tests (see Supplementary Table 1). This model estimates two different parameters for each item: item difficulty and item discrimination. Item difficulty describes how easy persons agree with the item. Item discrimination describes how well an item distinguishes persons with high and low ability.

To evaluate model fit, we used root mean square error of approximation (RMSEA), standardized root mean square residual (SRMSR), Tucker-Lewis index (TLI), comparative fit index (CFI). Good fit was defined as a RMSEA < 0.05, SRMSR ≤ 0.08 and > 0.9 for TLI and CFI [31]. Sample adjusted Bayesian information criterion (SABIC) and Akaike information criterion corrected (AIC_c) were examined as well, smaller values indicate a better model fit. We used infit and outfit statistics to evaluate item to model fit. The range of 0.5–1.5 is efficient for measurement, while the area between -1.9 and 1.9 describes reasonable predictability. Values ≤ -2 indicate data are too predictable [32]. Moreover, the relationship of choosing between answer categories of items and ability of patients is shown in a wright map. We used a test information curve to show the amount of ability measured over the ability range and to estimate the standard error. The GPCM was used to calculate the number of empirical distinguishable groups by the separation index [33]. To ensure answer behavior was not influenced by interviewers, differential item functioning (DIF) was assessed. The likelihood-ratio χ^2 test was used to detect DIF. McFadden's pseudo- R^2 and non-compensatory differential item functioning (NCDIF) were used as a measure of DIF magnitude.

Furthermore, floor and ceiling effects were defined as > 15% of answers in the lowest or highest answer category, respectively [34]. Additionally, item difficulty was investigated according to classical test theory (CTT), represented as item mean. For assessing CTT reliability, we used Cronbach's α (inner consistency). A value above 0.7 indicates acceptable reliability [35]. Indications for construct validity were gained through trait–trait correlations between the PAM® score and other questionnaires. It was expected that patient activation would be moderately negatively associated with self-related health status [36–38], moderately positively associated with selfefficacy [5], quality of life [9, 39, 40], and general mood [9, 41, 42], and weakly correlated with perceived social support [42, 43]. Correlations coefficients were judged as small if > 0.10, as medium if > 0.30 and as large if > 0.50 [44]. Group differences were evaluated by means of an Analysis of Variance. In case the overall comparison was significant, Tukey's HSD test was used for specific group differences.

Statistical analysis was performed using R studio version 4.1.1 [45] using the packages mirt [46] and lordif [47].

Results

Study participants

Overall, 707 patients were screened for eligibility. 81 Individuals (11%) were deemed ineligible due to inadequate German proficiency, cognitive incapacity, auditory impairments, or frailty 39 (6%) declined study participation. Six (1%) withdrew informed consent before the interview was started. Twelve interviews (2%) had to be stopped because patients were not suitable (e.g., hearing difficulties); 5 patients (1%) did the interview twice. 10 (1%) Interviews were incomplete, so 554 (78%) patients' data sets were available for analysis. N=321 (58%) of patients were male and median age was 69 (IQR 62–76). 317 (57%) Patients suffered from a macular edema due to diabetes, 224 (40%) due to retinal vein occlusion and 13 (2%) exhibited both types (see Table 1).

There were four different interviewers with interviewer MH doing 185 (33%), interviewer VW 159 (29%), interviewer AK 114 (21%) and interviewer JG 96 (17%) interviews, where each interviewer conducted as many interviews as feasible. One interview took about 20 min (IQR 16.4–24.8). The PAM-13D was finished in approximately four minutes (IQR 2.7–4.7).

PAM-13D results

The response "Not applicable" was chosen most frequently for item 4, occurring in 2% of all cases. The most frequent response category across all items was "Agree strongly", which was selected between 43% (item 9) and 88% (item 1) of patients, see Fig. 1. 10-38% of patients responded "Agree" over all items. "Disagree strongly" was the least chosen response category, for most items not exceeding 3% of answers. Mean response scores ranged from 3.1 (SD 0.9) for item 9 to 3.9 (SD 0.5) for item 1, see Table 2.

Study participants showed a mean PAM® score of 74.1 (SD 13.7), see Table 3. 19 (3%) Were in patient activation level 1, 37 (7%) level 2, 91 (16%) level 3, and 407 (73%) level 4.

	Total	PAM® score	<i>p</i> -value ^b
Age (years)	69.0 (62.0 - 76.0)		
Sex			0.018
Male	321 (58%)	72.9 ± 13.4	
Female	233 (42%)	75.7 ± 13.9	
Education			0.045 ^c
Basic education	411 (74%)	74.9 ± 13.8	
High school	68 (12%)	71.1 ± 14.0	
Higher education	75 (14%)	72.2 ± 12.3	
Working status			0.041 ^d
Working	66 (12%)	73.5 ± 12.7	
Other ^a	41 (7%)	69.1 ± 13.8	
Retired	447 (81%)	74.6 ± 13.7	
Monthly net income			0.172
<800€	72 (13%)	75.2 ± 13.1	
<1125€	91 (17%)	71.7 ± 14.5	
<1500€	112 (21%)	76.3 ± 13.7	
≤1950€	82 (15%)	74.0 ± 12.7	
> 1950€	184 (34%)	73.5 ± 13.5	
Missing	13 (2%)	74.4 ± 18.1	
BMI	25.8 (24.7–30.0)		
Type of macular edema	0.327		
Diabetic	317 (57%)	73.4 ± 13.7	
Retinal vein occlusion	224 (40%)	75.1 ± 13.7	
Both	13 (2%)	72.0 ± 11.3	
Comorbidity			0.224
Diabetes	129 (23%)	74.6 ± 12.6	
Hypertension	196 (35%)	75.1 ± 13.5	
Both	229 (41%)	72.9 ± 14.4	
Health status self-rated			<.001 ^e
Very good	61 (11%)	83.3 ± 11.1	
Good	236 (43%)	76.6 ± 11.7	
Moderate	222 (40%)	70.6 ± 13.8	
Bad	29 (5%)	60.8 ± 14.1	
Very bad	6 (1%)	72.7 ± 15.4	
Visual acuity tested		_	
Both eves	350 (63%)		
Only worse eye	160 (29%)		
No data	44 (8%)		
Visual acuity	· ·		LogMAR
Better eye	350		0.19 ± 0.20
Worse eye	510		0.42 ± 0.36

Data are presented as N(%), mean \pm SD or median (25th–75th percentiles)

BMI body mass index

^aCategory 'Other' includes jobless, studying and homemaker

^bEstimated difference in the PAM® score by an Analysis of Variance

^cNo significant difference between any specific groups in the post-hoc test (Tukey's HSD)

^dPost hoc test retired vs. other p = 0.032

^ePost hoc test p < 0.01 for very good vs. good; p < 0.001 for very good vs. moderate and bad, good vs. moderate and bad, moderate vs. bad. No missing data in PAM® scores

Outfit z

-0.5-0.7

-1.0

-1.3-2.2

-0.6-1.9

-1.6

-2.7

-2.1

-2.7

-3.3

-2.0



Fig. 1 Category answer frequencies for the PAM®

PAM® items	Order by IRT difference ^a	CTT ^b difficulty	IRT difficulty	IRT discrimi- nation	Infit mean squared	Infit z	Outfit mean squared
Item 01	1	3.9	-2.9	0.9	1.1	0.4	0.9
Item 02	2	3.8	-2.5	1.0	1.0	-0.1	0.9
Item 03	3	3.7	-2.3	0.8	1.0	0.0	0.9
Item 04	4	3.7	-1.9	0.8	1.0	-0.2	0.9
Item 05	7	3.4	-1.2	1.0	0.9	-1.4	0.9
Item 06	5	3.5	-1.7	0.5	1.0	-0.5	1.0
Item 07	10	3.3	-1.0	0.7	0.9	-1.4	0.9
Item 08	6	3.4	-1.3	0.7	1.0	-1.0	0.9
Item 09	13	3.1	-0.5	0.9	0.9	-2.4	0.9
Item 10	8	3.4	-1.2	0.8	0.9	-1.3	0.9
Item 11	9	3.4	-1.2	1.1	0.9	-1.5	0.8
Item 12	11	3.4	-0.9	1.3	0.9	-2.6	0.8
Item 13	12	3.3	-0.9	0.8	0.9	-1.6	0.9

^aItem sequence by item response theory (IRT) difficulty, ordered from easiest to hardest

^bCTT classical test theory. No missing data in PAM® items

Table 3 Questionnaire scores

	Mean	Standard devia- tion	Possible range
PAM® score	74.1	13.7	1-100
Self-efficacy	3.3	0.5	1–4
Quality of life	4.5	0.9	1–6
General mood	4.5	0.7	1–6

N = 554

Psychometric properties

Assumptions of IRT analysis were met sufficiently. The lowest two answer categories "Disagree strongly" and "Disagree" were chosen rarely, leading to estimation problems in the models. Therefore, these two categories were merged into one, named "Disagree strongly & Disagree". Firstly, we estimated a GPCM with all available data. Estimations of the ability level on the latent trait were used to impute former "Not applicable" answers (N = 23, 0.3% of all values). In the final model, the latent ability patient activation was estimated by the answer patterns of the 13 items of 554 participants, estimating difficulty and discrimination of every single item as well. Fit indices indicated a good model fit: RMSEA 0.062 (95% CI 0.052–0.072), SRMSR 0.064, TLI 0.905, CFI 0.921, SABIC 11,496.6 and AIC_c 11,458.1. The patient activation ability score calculated from the GPCM correlated 0.98 (p < 0.001) with the PAM® score. Items 9 and 12 showed bad in- and outfit values. Items 5, 10, 11 and 13 showed bad outfit values. Item difficulties ranged between -2.9 (item 1) to -0.5 (item 9), see Table 2. Item

Fig. 2 Wright map estimated by a GPCM. Top: distribution of the sample in activation scores in logits. Bottom: distributions of thresholds per item. The first threshold b1 describes the probability of choosing between "Disagree strongly & Disagree" or "Agree", the second threshold b2 between "Agree" or "Agree strongly" discrimination ranged from 0.5 to 1.3, with a mean of 0.9 (SD 0.2). Item characteristic curves are displayed in Supplementary Fig. 1.

Most of the sample had abilities between -2 and 1 logits, see Fig. 2. There is no item with a threshold for an ability higher than 0 logits. Total test information of the PAM-13D is 22.6, see Fig. 3. Most information is given in the ability area of -3 to 1 logits. Assessment of medium to highly activated patients is conflicted with more error than the lower ability range.

The GPCM estimated an empirical reliability of 0.75. Similarly, Cronbach's α was 0.75, indicating acceptable

Wright Map GPCM

Patient activation ability distribution



Fig. 3 Left side: PAM® test information (green line) and standard error (grey line) over the ability range in logits. Right side: PAM® empirical reliability over the ability range in logits. (Color figure online)



reliability. Using the GPCM, the estimation revealed two empirically distinguishable groups by the PAM-13D (e.g., low and high activation).

DIF was found for interviewer VW compared to all other interviewers for item 7 ($p < .001 R^2 = 0.028$, NCDIF = 0.123, Supplementary Fig. 2). This item was easier if administrated by interviewer VW.

PAM® score showed a Spearman correlation of -0.34 (p < .001) with health status, 0.54 (p < .001) with self-efficacy, 0.51 (p < .001) with quality of life, and 0.34 (p < .001) with general mood. For comparison, the patient activation ability score calculated from the GPCM correlated -0.35 (p < 0.001) with health status, 0.55 (p < .001) with self-efficacy, 0.52 (p < .001) with quality of life, and 0.34 (p < .001) with general mood.

Discussion

This study shows that the read-aloud PAM® survey demonstrates adequate measurement precision and construct validity to screen for patient activation in an everyday clinical setting. Patients with lower and medium activation are better differentiated than those with high activation. Additional research is needed to examine concerns regarding the structural validity. Since it takes about 4 min to administer the PAM® survey in an interview setting, an integration in clinical routine seems feasible.

Interviewer setting

Although these findings are promising, it must be considered that social desirability in an interviewer setting could influence the results. Participants could have over-reported their patient activation. As shown previously, compared to self-administration, interview responses are more positive and socially acceptable [20]. However, during the interview, most patients talked openly about their problems with selfcare in their everyday life. Another aspect that could lead to systematically different scores is a potential risk of interviewers influencing patients' responses [21], particularly when item explanations are provided. Thus, it is crucial to prioritize interviewer training, which should include standardized explanations of items. This study specifically aimed to closely examine whether interviewers had an impact on patients' answer behavior. More patients agreed to item 7 when interviewed by VW. Although only a very small effect was observed, this shows that proficient interviewer training is essential for the validity of the results. Since an influence of the interviewer on the answer behavior was only found for one item, we conclude that an objective assessment in an interview setting using the PAM-13D is possible. Previously, in an interview setting with individuals with vision loss [22], results regarding the PAM® survey were comparable to those found in other samples without vision loss [19, 48]. In this study patients showed high patient activation, with a mean PAM® score of 74.1 (SD 13.7) and 73% of participants being in PAM® level 4. This score is higher than scores previously found in Europe [49]. A comparison of activation scores in Europe revealed the German speaking sample as the most activated one [49]. Recently in Germany, the PAM® survey was administered in an interview setting with multimorbid patients. Similar scores as in this study with a mean of 76.1 (SD \pm 16.4) were reported for multimorbid patients [50]. Overall, these results indicate the feasibility of completing the questionnaire in an interview mode without being overly influenced by social desirability or interviewer bias.

Additionally, completing the PAM-13D in an interview setting created advantages: a conversation was partially developed and the interviewer got more information than the items asked for. Most of the patients answer more than just "Agree"—they do like to tell their story. For example, one patient had cookbooks for diabetics, but could not use recipes due to expensive ingredients. Actively managing her condition, she seeked guidance for preparing affordable, healthy meals. In a clinical setting, capturing additional information communicated by patients can aid in identifying the health challenges they are facing. A systematic review investigating the diagnosis of depression concluded that obtaining additional information from patients can aid in more specific management of health problems. The use of a self-administered questionnaire followed by an interview is recommended [51].

A further advantage of the interview setting was the opportunity for patients to inquire about the meanings of items they did not understand. Some items, especially item 7, required clarification through a standardized sentence.

Not only the interview setting, but also disease specific attributes may have an influence on patient activation. Many patients in this sample suffered not only from macular edema, but also diabetes or hypertension. The score of this study is higher compared to other samples with diabetes, where mean scores were 59 (SD \pm 12) [19], 57 (SD \pm 14) [52], 59 (SD \pm 10) [18]; or patients suffering from hypertension 61 (SD \pm 12) [53] and 61 (SD \pm 16) or a sample having both, diabetes and hypertension 60 (SD \pm 13) [54]. Therefore, differences in self-management tasks between different diseases must be considered when studying patient activation. Looking at severe chronic conditions, lung cancer patients showed similarly high activation scores as in this study [55]. Patients with life-threatening long-term illnesses (cardiac diseases or cancer) showed higher activation than other samples in the US [56]. Different vascular disease patient groups were investigated, the group with the most severe disease showed the most activation [17].

The psychometric properties of the read-aloud PAM[®]

The read-aloud PAM® survey shows adequate psychometric properties. Our findings indicate moderate reliability (0.75). While the items precisely assess patient activation in the lower and medium ability range, measurements in the high ability range are more error-prone. The reliability found in this study is lower than the reliability PAM-13D validation study (0.84) [5] and other European translations in (Denmark: 0.89, Italy: 0.88) [14, 15]. This study's lower reliability may result from the highly activated sample and the lack of difficult items. Moreover, construct validity is supported by expected associations with external criteria: higher activation was associated with better self-rated health, higher self-efficacy, higher quality of life and better general mood [6, 7, 9, 52]. Diabetics exhibiting higher activation levels had a better health status [19].

Investigating PAM-13D items, "Not applicable" was chosen rarely, mostly for item 4 "I know what each of my prescribed medications does"; consistent with other studies. Not everyone with a chronic disease necessarily takes medication. For patients with a macular edema due to retinal vein occlusion, this can be their first diagnosis, without a history of a chronic disease yet.

In this study, all items showed ceiling effects, aligning with similar findings in other European countries [14, 15]. Consistent with previous studies, "Disagree strongly" was chosen in $\leq 10\%$ of the responses [5, 15] necessitating merging with the "Disagree" category for analysis [8, 48].

During the analysis, six items showed a poor model fit, suggesting influences beyond patient activation on response behavior. These items (5, 9, 10, 11, 12 and 13) displayed highest difficulties in this study. To our knowledge, this was not found in other studies [6, 18, 22], but, they did not report the *z*-outfit, where the misfit occurred in our study.

According to the original questionnaire design, item difficulty increases sequentially, while items align with specific activation levels. This applied only to the first four items in this study. Similar results were found in the PAM-13D [5], other translations [14, 49] and a diabetic sample [18]. These findings, including ours, might result from specific disease and cultural factors. Although the original goal was to increase item difficulty over the course of the questionnaire, its absence in many studies is no problem for the interpretation of the overall PAM® score. Caution is advised when evaluating health behavior on a few items, as they may not measure the intended patient activation level. Moreover, the classification into four different activation levels could not be replicated in this study. Our analysis estimated how many groups the questionnaire distinguishes empirically. It suggested categorizing patients into two groups instead of four.

Limitations and future recommendations

As in every survey, the question arises as to whether there is a sample bias. Uncertainty remains whether interviewed patients from one center are representative for patients with macular edema. It should be emphasized positively that almost all recurrent patients were interviewed during the study period of two years. A further concern is a possible selection bias. While this effect is described in many studies (e.g. [16]), selection bias in this study can only be small, since 94% of the potentially eligible patients could be included.

This study demonstrates administrating questionnaires to a population, unable to complete them by themselves, delivers reliable and valid results. It is possible to overcome barriers due to patients' characteristics to gain more insight about patients' health behavior. Therefore, in clinical practice and future research, an atmosphere should be established where the patient feels understood and his thoughts and feelings welcome to reduce social desirability. Since the interviewer situation has an influence on patients, a standardized procedure is necessary including standardized explanations for items.

Conclusions

This study suggests the PAM® survey can be read-aloud in everyday clinical settings to assess patient activation of patients with chronic diseases. All items showed ceiling effects and these were rarely disagreed. This raises the question of whether it is necessary for all items to be answered with four response categories. Due to the interview setting, we recognized that some items are not well understood. Proficient interviewer training is essential to ensure consistent explanation for patients. While the PAM® survey effectively captures activation in lower to middle-activated patients, it falls short for highly activated patients. Therefore, we recommend using it as a screening tool rather than a diagnostic measure. To enhance the ability to discern highly activated patients and better understand the challenges they face, we propose rephrasing existing items to increase difficulty or incorporating new, specific items. Despite limitations, PAM® survey is valuable in identifying low activation, creating the opportunity to encourage skills, knowledge, and confidence in the management of a chronic disease.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s11136-024-03614-2.

Acknowledgements We thank all the staff members at the Department of Ophthalmology for making this study possible and the patients for their cooperation. Moreover, we are grateful for the Doctoral School of the Medical University of Graz taking over the article-processing charges.

Author contributions We confirm that all authors were involved in the design and implementation of the study, or acquisition of data, or analysis and interpretation of data. All authors contributed and approved to the manuscript. MH, AA, AB and AW took part in planning the study; MW, SS and MM recruited patients; MH did data assessment; MH, AA and KBF data analysis; MH main part writing manuscript; all authors contributed to the manuscript.

Funding Open access funding provided by Medical University of Graz. This manuscript represents a part of a project that did not receive any specific grant from funding agencies in the public, commercial, or nonprofit sectors and was sponsored by the Medical University of Graz. Article-processing charges are kindly covered by the Doctoral School of the Medical University of Graz.

Data availability The participants of this study did not give written consent for their data to be shared publicly, so due to the sensitive nature of the research the data can not be made available.

Declarations

Conflict of interest The authors declare that they have no competing interests.

Ethical approval The Ethical Committee of the Medical University of Graz approved the study (32-101 ex 19/20) in accordance with the Declaration of Helsinki.

Informed consent Participants provided written informed consent to participate in this study and were able to opt out at any moment.

Consent for publication Not applicable. All personal data are deidentified.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- 1. World Health Organization. (2022). World health statistics 2022: Monitoring health for the SDGs, sustainable development goals. World Health Organization.
- Hibbard, J. H., Stockard, J., Mahoney, E. R., & Tusler, M. (2004). Development of the Patient Activation Measure (PAM): Conceptualizing and measuring activation in patients and consumers. *Health Services Research*, 39, 1005–1026.
- Mosen, D. M., Schmittdiel, J., Hibbard, J., Sobel, D., Remmers, C., & Bellows, J. (2007). Is patient activation associated with outcomes of care for adults with chronic conditions? *Journal of Ambulatory Care Management*, 30, 21–29.
- Dixon, A., Hibbard, J., & Tusler, M. (2009). How do people with different levels of activation self-manage their chronic conditions? *The Patient: Patient-Centered Outcomes Research*, 2, 257–268.
- Brenk-Franz, K., Hibbard, J. H., Herrmann, W. J., Freund, T., Szecsenyi, J., Djalali, S., Steurer-Stey, C., Sönnichsen, A., Tiesler, F., Storch, M., Schneider, N., Gensichen, J. (2013). Validation of the German version of the Patient Activation Measure 13 (PAM13-D) in an international multicentre study of primary care patients. *PLoS ONE*, *8*, e74786.
- Magnezi, R., Glasser, S., Shalev, H., Sheiber, A., & Reuveni, H. (2014). Patient activation, depression and quality of life. *Patient Education and Counselling*, *94*, 432–437.
- Moreno-Chico, C., González-de Paz, L., Monforte-Royo, C., Arrighi, E., Navarro-Rubio, M. D., & Gallart, F.-P. (2017). Adaptation to European Spanish and psychometric properties of the Patient Activation Measure 13 in patients with chronic diseases. *Family Practice*, 34, 627–634.
- Packer, T. L., Kephart, G., Ghahari, S., Audulv, Å., Versnel, J., & Warner, G. (2015). The Patient Activation Measure: A validation study in a neurological population. *Quality of Life Research*, 24, 1587–1596.
- Eyles, J. P., Ferreira, M., Mills, K., Lucas, B. R., Robbins, S. R., Williams, M., Lee, H., Appleton, S., Hunter. D. J. (2020). Is the Patient Activation Measure a valid measure of osteoarthritis self-management attitudes and capabilities? Results of a Rasch analysis. *Health and Quality of Life Outcomes*, 18, 121.
- Chen, W.-H., Lenderking, W., Jin, Y., Wyrwich, K. W., Gelhorn, H., & Revicki, D. A. (2014). Is Rasch model analysis applicable in small sample size pilot studies for assessing item characteristics?

- Moljord, I. E. O., Lara-Cabrera, M. L., Perestelo-Pérez, L., Rivero-Santana, A., Eriksen, L., & Linaker, O. M. (2015). Psychometric properties of the Patient Activation Measure-13 among out-patients waiting for mental health treatment: A validation study in Norway. *Patient Education and Counseling*, 98, 1410–1417.
- 12. Hibbard, J. H., Mahoney, E. R., Stock, R., & Tusler, M. (2007). Do increases in patient activation result in improved self-management behaviors? *Health Services Research*, *42*, 1443–1463.
- Hibbard, J. H., Mahoney, E. R., Stockard, J., & Tusler, M. (2005). Development and testing of a short form of the Patient Activation Measure. *Health Services Research*, 40, 1918–1930.
- Graffigna, G., Barello, S., Bonanomi, A., Lozza, E., & Hibbard, J. (2015). Measuring patient activation in Italy: Translation, adaptation and validation of the Italian version of the Patient Activation Measure 13 (PAM13-I). *BMC Medical Informatics and Decision Making*, 15, 109.
- Maindal, H. T., Sokolowski, I., & Vedsted, P. (2009). Translation, adaptation and validation of the American short form Patient Activation Measure (PAM13) in a Danish version. *BMC Public Health*, 9, 209.
- Skelly, A., Taylor, N., Fasser, C., Malkowski, J.-P., Goswamy, P., & Downey, L. (2022). Patient preferences in the management of wet age-related macular degeneration: A conjoint analysis. *Advances in Therapy*. https://doi.org/10.1007/s12325-022-02248-5
- Humphries, M. D., Welch, P., Hasegawa, J., & Mell, M. W. (2021). Correlation of patient activation measure level with patient characteristics and type of vascular disease. *Annals of Vascular Surgery*, 73, 55–61.
- Laranjo, L., Dias, V., Nunes, C., Paiva, D., & Mahoney, B. (2018). Translation and validation of the Patient Activation Measure in Portuguese people with Type 2 diabetes mellitus. *Acta Médica Portuguesa*, 31, 382.
- van Vugt, H. A., Boels, A. M., de Weerdt, I., de Koning, E. J. P., & Rutten, G. E. H. M. (2018). Patient activation in individuals with Type 2 diabetes mellitus: Associated factors and the role of insulin. *Patient Preference and Adherence*, 13, 73–81.
- Bowling, A. (2005). Mode of questionnaire administration can have serious effects on data quality. *Journal of Public Health*, 27, 281–291.
- 21. Cook, C. (2010). Mode of administration bias. *Journal of Manual and Manipulative Therapy*, *18*, 61–63.
- 22. Morse, A. R., & Seiple, W. (2021). Activation in individuals with vision loss. *Journal of Health Psychology*, *26*, 2603–2612.
- Yau, J. W. Y., Lee, P., Wong, T. Y., Best, J., & Jenkins, A. (2008). Retinal vein occlusion: An approach to diagnosis, systemic risk factors and management. *Internal Medical Journal*, 38, 904–910.
- Im, J. H. B., Jin, Y.-P., Chow, R., & Yan, P. (2022). Prevalence of diabetic macular edema based on optical coherence tomography in people with diabetes: A systematic review and meta-analysis. *Surveys in Ophthalmology*, 67, 1244–1251.
- Institute of Medical Informatics, Statistics and Documentation, Medical University of Graz, Austria. (2023). Retrieved September 15, 2022, from https://imi.medunigraz.at/en/services#c44617
- Startseite LimeSurvey einfache Online-Umfragen. (2023). Retrieved September 11, 2022, from https://www.limesurvey.org/ de/
- 27. Dalbert, C. (1992). *HSWBS-Habituelle subjektive Wohlbefindens-skala*. ZPID Testarchiv der Universität Trier.
- 28. Schwarzer, R., & Jerusalem, M. (1999). Skalen zur erfassung von Lehrer-und schülermerkmalen. Institut für Psychologie.
- De Bruin, A. (1996). Health interview surveys: Towards international harmonization of methods and instruments. WHO Regional Publications, European Series, No. 58. ERIC.

- Cappelleri, J. C., Lundy, J. J., & Hays, R. D. (2014). Overview of classical test theory and item response theory for the quantitative assessment of items in developing patient-reported outcomes measures. *Clinical Therapeutics*, *36*, 648–662.
- Hu, L., & Bentler, P. M. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural Equation Modeling: A Multidisciplinary Journal*, 6, 1–55.
- 32. Linacre, J. M. (2002). What do Infit and Outfit, mean-square and standardized mean? *Rasch Measurement Transactions*, *16*(2), 878.
- 33. Wright, B. D., & Masters, G. N. (1982). *Rating scale analysis*. Mesa Press.
- McHorney, C. A., & Tarlov, A. R. (1995). Individual-patient monitoring in clinical practice: Are available health status surveys adequate? *Quality of Life Research*, 4, 293–307.
- Gliem, J. A., & Gliem, R. R. (2003). Calculating, interpreting, and reporting Cronbach's alpha reliability coefficient for Likerttype scales. In: *Midwest research-to-practice conference in adult, continuing, and community education*, 2003.
- Skolasky, R. L., Green, A. F., Scharfstein, D., Boult, C., Reider, L., & Wegener, S. T. (2011). Psychometric properties of the Patient Activation Measure among multimorbid older adults: Psychometric properties of the PAM. *Health Services Research*, 46, 457–478.
- Haj, O., Lipkin, M., Kopylov, U., Sigalit, S., & Magnezi, R. (2022). Patient activation and its association with health indices among patients with inflammatory bowel disease. *Therapeutic Advances in Gastroenterology*, 15, 175628482211287.
- Tusa, N., Kautiainen, H., Elfving, P., Sinikallio, S., & Mäntyselkä, P. (2020). Relationship between patient activation measurement and self-rated health in patients with chronic diseases. *BMC Family Practice*, 21, 225.
- 39. Magadi, W., Lightfoot, C. J., Memory, K. E., Santhakumaran, S., van der Veer, S. N., Thomas, N., Gair, R., Smith, A. C. (2022). Patient activation and its association with symptom burden and quality of life across the spectrum of chronic kidney disease stages in England. *BMC Nephrology*, 23, 45.
- Hendriks, S. H., Hartog, L. C., Groenier, K. H., Maas, A. H. E. M., Van Hateren, K. J. J., Kleefstra, N., Bilo, H. J. G. (2016). Patient activation in Type 2 diabetes: Does it differ between men and women? *Journal of Diabetes Research*, 2016, 1–8.
- Magnezi, R., & Glasser, S. (2014). Psychometric properties of the Hebrew translation of the Patient Activation Measure (PAM-13). *PLoS ONE*, 9, e113391.
- Gleason, K. T., Tanner, E. K., Boyd, C. M., Saczynski, J. S., & Szanton, S. L. (2016). Factors associated with patient activation in an older adult population with functional difficulties. *Patient Education and Counseling*, *99*, 1421–1426.
- 43. Bos-Touwen, I., Schuurmans, M., Monninkhof, E. M., Korpershoek, Y., Spruit-Bentvelzen, L., Ertugrul-van der Graaf, I., de Wit, N., Trappenburg, J. (2015). Patient and disease characteristics associated with activation for self-management in patients with diabetes, chronic obstructive pulmonary disease, chronic heart failure and chronic renal disease: A cross-sectional survey study. *PLoS ONE*, 10, e0126400.
- Cohen, J. (2009). Statistical power analysis for the behavioral sciences (2nd ed., reprint). Psychology Press.
- RStudio Team. (2021). RStudio: Integrated development environment for R. RStudio, PBC. http://www.rstudio.com/
- Philip Chalmers, R. (2012). mirt: A multidimensional item response theory package for the R environment. *Journal of Statistical Software*, 48(6), 1–29. https://doi.org/10.18637/jss.v048. i06
- Choi, S. W. with contributions from Gibbons, L. E. and Crane, P. K. (2016). lordif: Logistic ordinal regression differential item

functioning using IRT. R package version 0.3-3. https://CRAN.Rproject.org/package=lordif

- Hellström, A., Tessma, M. K., Flink, M., Dahlgren, A., Schildmeijer, K., Ekstedt, M. (2019). Validation of the Patient Activation Measure in patients at discharge from hospitals and at distance from hospital care in Sweden. *BMC Public Health*, 19, 1701.
- Rademakers, J., Maindal, H. T., Steinsbekk, A., Gensichen, J., Brenk-Franz, K., & Hendriks, M. (2016). Patient activation in Europe: An international comparison of psychometric properties and patients' scores on the short form Patient Activation Measure (PAM-13). BMC Health Services Research, 16, 570.
- Breckner, A., Glassen, K., Schulze, J., Lühmann, D., Schaefer, I., Szecsenyi, J., Scherer, M., Wensing, M. (2022). Experiences of patients with multimorbidity with primary care and the association with patient activation: A cross-sectional study in Germany. *British Medical Journal Open*, *12*, e059100.
- Costantini, L., Pasquarella, C., Odone, A., Colucci, M. E., Costanza, A., Serafini, G., Aguglia, A., Murri, M. B., Brakoulias, V., Amore, M., Ghaemi, S. N., Amerio, A., (2021). Screening for depression in primary care with Patient Health Questionnaire-9 (PHQ-9): A systematic review. *Journal of Affective Disorders*, 279, 473–483.
- Hendriks, M., & Rademakers, J. (2014). Relationships between patient activation, disease-specific knowledge and health outcomes among people with diabetes; a survey study. *BMC Health Services Research*, 14, 393.

- Lee, C., & Horowitz, C. R. (2022). Association between APOL1 risk testing in primary care settings and patient engagement among Black patients with hypertension. *Research Square*. https:// doi.org/10.21203/rs.3.rs-1180383/v1
- Zeng, H., Jiang, R., Zhou, M., Wu, L., Tian, B., Zhang, Y., Qu, F. (2019). Measuring patient activation in Chinese patients with hypertension and/or diabetes: Reliability and validity of the PAM13. *Journal of International Medical Research*, 47, 5967–5976.
- 55. Sun, V., Raz, D. J., Ruel, N., Chang, W., Erhunmwunsee, L., Reckamp, K., Tiep, B., Ferrell, B., McCorkle, R., Kim, J. Y. (2017). A multimedia self-management intervention to prepare cancer patients and family caregivers for lung surgery and postoperative recovery. *Clinical Lung Cancer*, 18, e151–e159.
- Jerofke, T., Weiss, M., & Yakusheva, O. (2014). Patient perceptions of patient-empowering nurse behaviours, patient activation and functional health status in postsurgical patients with life-threatening long-term illnesses. *Journal of Advanced Nursing*, 70, 1310–1322.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.